Author’s response to reviews

Title: Co-occurrence of risky health behaviours: results from the Pre-Empt study.

Authors:

Elizabeth Randell (randelle@cardiff.ac.uk)
Timothy Pickles (pickleste@cardiff.ac.uk)
Sharon A Simpson (sharon.simpson@glasgow.ac.uk)
Clio Spanou (clio.spanou@beds.ac.uk)
Jim McCambridge (jim.mccambridge@york.ac.uk)
Kerenza Hood (hoodk1@cardiff.ac.uk)
Christopher C Butler (butlercc@cardiff.ac.uk)

Version: 2 Date: 14 April 2015

Author’s response to reviews:

Following initial assessment, a request was made to include abbreviations, a statement of ethics and to move the tables from additional files to the main manuscript. This has all been done. A new version of the main manuscript is now being uploaded and the separate additional files containing table 1-4 will be removed.